Author's response to reviews

Title: Fetal loss and maternal serum levels of 2,2',4,4',5,5'-hexachlorbiphenyl (CB-153) and 1,1-dichloro-2,2-bis(p-chlorophenyl)ethylene (p,p'-DDE) exposure: a cohort study in Greenland and two European populations

Authors:

Gunnar Toft (guntof@rm.dk)
Ane M Thulstrup (anemt@rm.dk)
Bo A Jönsson (Bo_A.Jonsson@med.lu.se)
Henning S Pedersen (HSP@MIL.AU.DK)
Jan K Ludwicki (kludwicki@pzh.gov.pl)
Valentyna Zvezday (deemusik@yandex.ru)
Jens P Bonde (jpb@bbh.regionh.dk)

Version: 5 Date: 19 April 2010

Author's response to reviews:

I have made formatted the manuscript as suggested by the Environmental Health Editorial team.